Author's response to reviews

Title: Unnecessary Use of Fluoroquinolone Antibiotics in Hospitalized Patients

Authors:

Nicole L Werner (Nicole.Werner@case.edu)
Michelle T Hecker (MHecker@metrohealth.org)
Ajay K Sethi (aksethi@wisc.edu)
Curtis J Donskey (curtisd123@yahoo.com)

Version: 4 Date: 27 May 2011

Author's response to reviews:

May 27, 2011

Dear Editor,

We would like to submit a revised version of our manuscript entitled “Unnecessary use of fluoroquinolone antibiotics in hospitalized patients” for consideration for publication in BioMed Central Infectious Diseases. Our response the editorial request is below.

Thank you,

Curtis Donskey, M.D.
Cleveland VA Medical Center
Geriatric Research, Education and Clinical Center
10701 East Boulevard, Cleveland, Ohio

Editorial request: 1) Please can you include the aims of your study within the Background section of your Abstract.

Response: We have added the aims of the study in the abstract on lines 25 to 28 of the revised manuscript (highlighted in red text in the revised manuscript).